Author’s response to reviews

Title: Does External Beam Radiation Boost to Pelvic Lymph Nodes Improve Outcomes in Patients with Locally Advanced Cervical Cancer?

Authors:

Caryn Wujanto (caryn.wujanto@gmail.com; caryn.wujanto@mohh.com.sg)
Bok Ai Choo (bok_ai_choo@nuhs.edu.sg)
David Tan (David_sp_tan@nuhs.edu.sg)
Arunachalam Ilancheran (ilancheran_arunachalam@nuhs.edu.sg)
Joseph Ng (Joseph_ng@nuhs.edu.sg)
Jeffrey Low (Jeffrey_low@nuhs.edu.sg)
Liang Shen (medsl@nus.edu.sg)
Johann Tang (Johann_tang@nuhs.edu.sg)
Vicky Koh (Vicky_koh@nuhs.edu.sg)

Version: 4 Date: 05 Apr 2019

Author’s response to reviews:

Dear Editor and Reviewers,

Thank you very much for making time to review our manuscript. Our responses are as follows:

1. We noticed that the corresponding author email address provided on your title page is different to the one you indicated in the editorial submission. Please amend one so that they match.

Response: Apologies for the oversight. We will be using the corresponding author email address as provided in the title page and will amend the editorial submission.
2. If consent was not required, please provide details of why this was waived and the institution/committee that approved the waiver.

Response: As this was a retrospective study, consent was waived as all data used in the analysis were fully anonymised. We carried out this study following approval from the National Healthcare Group Domain Specific Review Board, Singapore.